Author's response to reviews

Title: The relationship between acculturation strategies and depressive and anxiety disorders in Turkish migrants in the Netherlands

Authors:

Burçin Ünlü Ince (b.unlu@vu.nl)
Thijs Fassaert (tfassaert@ggd.amsterdam.nl)
Matty A.S. de Wit (MdWit@ggd.amsterdam.nl)
Pim Cuijpers (p.cuijpers@vu.nl)
Jan Smit (JH.Smit@ggzingeest.nl)
Jeroen Ruwaard (jruwaard@me.com)
Heleen Riper (h.riper@vu.nl)

Version: 4 Date: 23 August 2014

Author's response to reviews: see over
23/08/2013

Dear dr. Powers,

Thank you for accepting our manuscript entitled "The relationship between acculturation strategies and depressive and anxiety disorders in Turkish migrants in the Netherlands" (MS: 4549027891227590).

As instructed, we have carefully addressed the editorial request. Revisions in text are shown with the Track Changes feature. Below you can find our reply to the request. Please find attached a revised version of our MS.

We hope that the revisions and our accompanying responses will be sufficient.

We look forward to the publication.

Yours sincerely,

Burçin Ünlü Ince
Response to the editorial request

Comment: Please can you include funding sources for this study and for each author in your acknowledgement section.

Response: The funding for the data-collection was obtained by the Municipality of Amsterdam, the mental health care institutions Arkin (formerly JellinekMentrum and AMC de Meren) and GGZinGeest (formerly Stichting Buitenamstel Geestgronden). None of the authors obtained extra funding for the academic collaboration.

We incorporated this in the acknowledgements.